Author's response to reviews

Title: Nine-year incident diabetes is predicted by fatty liver indices: the French D.E.S.I.R. study

Authors:

Beverley Balkau (beverley.balkau@inserm.fr)
Celine Lange (celine.lange@inserm.fr)
Sylviane Vol (sylviane.vol@irsa.asso.fr)
Frederic Fumeron (frederic.fumeron@inserm.fr)
Fabrice Bonnet (fabrice.bonnet@chu-rennes.fr)
the D.E.S.I.R. Study Group (not@valid.com)

Version: 6 Date: 3 June 2010

Author's response to reviews:

I had problems with the site resubmitting several days - but had no response to my query